Author's response to reviews

Title: Enhancement of antimicrobial activities of whole and sub-fractionated white tea by addition of copper (II) sulphate and vitamin C against Staphylococcus aureus; a mechanistic approach

Authors:

Andrew C Holloway (A.Holloway@kingston.ac.uk)
Simon W J Gould (S.Gould@kingston.ac.uk)
Mark D Fielder (M.Fielder@kingston.ac.uk)
Declan P Naughton (D.Naughton@kingston.ac.uk)
Alison F Kelly (A.Kelly@kingston.ac.uk)

Version: 3 Date: 28 October 2011

Author's response to reviews: see over
Dear Sir/Madam,

The authors would like to thank the editorial board for accepting the revised manuscript entitled “Enhancement of antimicrobial activities of whole and sub-fractionated white tea by addition of copper (II) sulphate and vitamin C against Staphylococcus aureus; a mechanistic approach” for publication as a research article in BMC Complimentary and Alternative Medicine.

The major and minor revisions requested have been completed and the final document read through for accuracy. It is hoped that all revisions have been done to the journals specifications.

Please do not hesitate to contact me for any further information you may require.

Yours faithfully,

Alison Kelly